

EUROPEAN CENTRAL BANK

WORKING PAPER SERIES NO. 426 / JANUARY 2005

ECB-CFS RESEARCH NETWORK ON CAPITAL MARKETS AND FINANCIAL INTEGRATION IN EUROPE

SECURITY FUNGIBILITY AND THE COST OF CAPITAL

EVIDENCE FROM GLOBAL BONDS

by Darius P. Miller and John J. Puthenpurackal



WORKING PAPER SERIES

NO. 426 / JANUARY 2005

ECB-CFS RESEARCH NETWORK ON CAPITAL MARKETS AND FINANCIAL INTEGRATION IN EUROPE

SECURITY FUNGIBILITY AND THE COST OF CAPITAL EVIDENCE FROM GLOBAL BONDS '

> by Darius P. Miller² and John J. Puthenpurackal³

This paper can be downloaded without charge from http://www.ecb.int or from the Social Science Research Network electronic library at http://ssrn.com/abstract_id=634021.







publications will feature a motif taken from the €50 banknote.



I We thank Utpal Bhattacharya, Charles Cuny, Vihang Errunza, Donald Fraser, Robert Jennings, Jonathan Karpoff (the editor), Leora Klapper, Heike Reichelt (World Bank), Rene Stulz, an anonymous referee as well as seminar participants at American University, the Federal Reserve Board of Governors, Texas A&M University, the Swedish School of Economics, the 2002 Financial Management Association and the 2002 European Financial Management Association for helpful suggestions. We especially thank Susan Chaplinsky for suggestions that have improved the quality of the paper. Jerry Martin, Deepa Raghavan, Rina Ray and Peter Tran (IFC) graciously offered programming and data collection assistance. Darius Miller thanks DaimlerChrysler and John Puthenpurackal acknowledges the Gardner Fellowship, for financial support.

2 Kelley School of Business, Indiana University, 1309 E.Tenth Street, Bloomington, IN 47405, USA; Phone: 812.855.3395; e-mail: damiller@indiana.edu

3 Finance Department, College of Business, Ohio University, 212 Copeland Hall, Athens, OH 45701, USA; Phone: 740.597.1870; e-mail: puthenpu@ohio.edu

ECB-CFS Research Network on "Capital Markets and Financial Integration in Europe"

This paper is part of the research conducted under the ECB-CFS Research Network on "Capital Markets and Financial Integration in Europe". The Network aims at stimulating top-level and policy-relevant research, significantly contributing to the understanding of the current and future structure and integration of the financial system in Europe and its international linkages with the United States and Japan. After two years of work, the ECB Working Paper Series is issuing a selection of papers from the Network. This selection is covering the priority areas "European bond markets", "European securities settlement systems", "Bank competition and the geographical scope of banking activities", "international portfolio choices and asset market linkages" and "start-up financing markets". It also covers papers addressing the impact of the euro on financing structures and the cost of capital.

The Network brings together researchers from academia and from policy institutions. It has been guided by a Steering Committee composed of Franklin Allen (University of Pennsylvania), Giancarlo Corsetti (European University Institute), Jean-Pierre Danthine (University of Lausanne), Vítor Gaspar (ECB), Philipp Hartmann (ECB), Jan Pieter Krahnen (Center for Financial Studies), Marco Pagano (University of Napoli "Federico II") and Axel Weber (CFS). Mario Roberto Billi, Bernd Kaltenhäuser (both CFS), Simone Manganelli and Cyril Monnet (both ECB) supported the Steering Committee in its work. Jutta Heeg (CFS) and Sabine Wiedemann (ECB) provided administrative assistance in collaboration with staff of National Central Banks acting as hosts of Network events. Further information about the Network can be found at http://www.eu-financial-system.org.

The joint ECB-CFS Research Network on "Capital Markets and Financial Integration in Europe" aims at promoting high quality research. The Network as such does not express any views, nor takes any positions. Therefore any opinions expressed in documents made available through the Network (including its web site) or during its workshops and conferences are the respective authors' own and do not necessarily reflect views of the ECB, the Eurosystem or CFS.

© European Central Bank, 2005

Address Kaiserstrasse 29 60311 Frankfurt am Main, Germany

Postal address Postfach 16 03 19 60066 Frankfurt am Main, Germany

Telephone +49 69 1344 0

Internet http://www.ecb.int

Fax +49 69 1344 6000

Telex 411 144 ecb d

All rights reserved.

Reproduction for educational and noncommercial purposes is permitted provided that the source is acknowledged.

The views expressed in this paper do not necessarily reflect those of the European Central Bank.

The statement of purpose for the ECB Working Paper Series is available from the ECB website, http://www.ecb.int.

ISSN 1561-0810 (print) ISSN 1725-2806 (online)

CONTENTS

Ab	stract	4
No	n-technical summary	5
1.	Introduction	7
2.	A primer on global bonds	9
3.	Potential benefits and costs of globally	
	fungible securities	11
	3.1 Liquidity	11
	3.1.1 Cross-border transactions cost	12
	3.2 Investor base and market conditions	12
	3.3 Issuing costs (gross spreads)	13
	3.4 Costs of global bond issuance	13
4.	Issue and issuer characteristics	14
	4.1 Data selection and issue characteristics	14
	4.2 Characteristics of global and non-global issuers	16
5	Testing for benefits of global bonds	17
	5.1 Comparing yields of global bonds and	
	domestic bonds issued by U.S. firms	17
	5.1.1 Controlling for potential lack of	
	independence of bond issues in	
	multi-tranche issuances	18
	5.1.2 Controlling for potential endogeneity in the decision to issue globally	18
	5.1.3 Treatment effects model using	
	maximum likelihood estimator	19
	5.1.4 Relative borrowing costs of global and domestic bonds of U.S. firms	22
	5.1.5 Relative issuing costs	
	(gross spreads) of global and	
	domestic bonds of U.S. firms	23
	5.2 Robustness tests	24
6.	Stock price reaction analyses	26
7.	Additional tests	27
	Conclusion	29
	ferences	31
1	pendix A	34
	bles	35
Eu	ropean Central Bank working paper series	41

Abstract

This paper examines the potential benefits of security fungibility by conducting the first comprehensive analysis of Global bonds. Unlike other debt securities, Global bonds' fungibility allows them to be placed simultaneously in bond markets around the world; they trade, clear and settle efficiently within as well as across markets. We test the impact of issuing these securities on firms' cost of capital, issuing costs, liquidity and shareholder wealth. Using a sample of 230 Global bond issues by 94 companies from the U.S. and abroad over the period 1996-2003, we find that firms are able to lower their cost of (debt) capital by issuing these fungible securities. We also document that the stock price reaction to the announcement of Global bond issues over the same time period are associated with insignificant changes in shareholder wealth.

JEL classification: F3, G1, G3

Keywords: Global bonds; security fungibility; cost of capital; international capital raising

Non-technical summary

This paper examines the potential benefits of security fungibility by conducting the first comprehensive analysis of Global bonds. Global bonds are debt instruments that are sold simultaneously in multiple markets at the same offer price. They are fully fungible in that the identical instrument trades within each market as well as between markets without restrictions. Global bonds are similar to domestic public corporate debt except that they are placed with a diverse international investor base that can easily trade them within and across markets around the world. To accomplish this, new global custody, clearing and settlement procedures have been created to streamline cross-market trading. We examine potential benefits of these fungible securities by studying their impact on firms' cost of capital, issuing costs (gross spreads) and shareholder wealth.

Using a sample of 230 Global bond issues from 94 companies from the U.S. and abroad over the 1996-2003 time period, we find that both the borrowing costs and issuing costs (underwriting spreads) of Global bonds are significantly lower than those of comparable domestic bonds and Eurobonds. Specifically, our results suggest that firms that issue globally are able to lower their borrowing costs by approximately 20 basis points relative to non-Global bonds, ceteris paribus. This result is robust to a number of tests that attempt to control for alternative benchmarks, potential endogeneity in the decision to issue globally as well as issue size and issuer related differences. Our results are consistent with the hypothesis that firms are able to lower their cost of (debt) capital by issuing fungible debt securities that are more liquid, have lower cross-border transactions costs, have longer trading hours and can access a wider investor base.

We also document that the stock price reaction to the announcement of Global bond issuance is positive and significant, while comparable domestic and Eurobond issues by U.S. firms over the same time period are associated with insignificant changes in shareholder wealth. These results suggest that the benefit of global issuance is not being driven by global issuers exploiting temporary differences in the Eurodollar-U.S. interest rate. Overall, our findings suggest that the issuance of globally tradeable securities is associated with significant benefits.

Our study of Global bonds provides, to the best of our knowledge, the first large sample test of the impact of the integration of cross-border clearing and settlement systems, a process whose benefits are currently under debate. Our finding that Global bonds' fungibility lowers borrowing costs provides evidence that the wave of consolidation and integration of securities clearing and settlement organizations and the call for the creation of a global clearing and settlement solution would indeed have benefits to firms' capital raising activities.

1. Introduction

Although global markets are becoming increasingly integrated, it is still the case that stock and bond trading around the world is largely confined to the market where the security was first issued. A share of IBM that trades on the NYSE is not able to trade on the Tokyo, Frankfurt, or any other exchange. IBM's public debt securities are also cumbersome to trade across borders, since registration, clearing, and settlement are all different for U.S., Euro, and Asian bond markets. In contrast, what if a firm's securities were engineered to achieve maximum tradeability, thereby facilitating placement with an international investor base and easy trading within and across multiple markets around the world? Would firms be able to raise more capital at better rates? In this paper, we examine these questions using a sample of Global bonds, the first widely issued securities that are designed for maximum tradeability.

Global bonds are debt instruments that are sold simultaneously in multiple markets at the same offer price. They are fully fungible in that the identical instrument trades within each market as well as between markets without restrictions. ¹ Global bonds are similar to domestic public corporate debt except that they are placed with a diverse international investor base that can easily trade them within and across markets around the world. To accomplish this, new global custody, clearing and settlement procedures were created to streamline cross-market trading. While the first issuance of a Global bond by a non-financial firm was in 1992, over \$300 billion has been raised by corporations since their introduction.

Using a sample of 230 Global bond issues from 94 companies from the U.S. and abroad over the 1996-2003 time period, we find that both the borrowing costs and issuing costs (underwriting spreads) of Global bonds are significantly lower than those of comparable domestic bonds and Eurobonds. Therefore, our findings suggest that issuing Global bonds, which have greater liquidity and lower issuing costs than non-Global bonds, reduces the cost of debt capital. We also document that the stock price reaction to the announcement of Global bond issuance is positive and significant, while comparable

¹ All financial contracts with identical terms are not necessarily fully fungible. For example, the stock certificate that represents one share of IBM on the NYSE cannot be used to trade for the certificate that represents one share of IBM on the Tokyo stock exchange. See Gagnon and Karolyi (2003) for additional details on international security fungibility in the context of equities.

domestic and Eurobond issues over the same time period are associated with insignificant changes in shareholder wealth. Overall, our findings suggest that issuing globally tradeable securities is associated with economically significant benefits.

This study contributes to the literature in several ways. First, we examine the potential benefits of a financial instrument that is becoming the customary way for corporations, countries, and supranationals to raise large amounts of funds.² Our study of Global bonds also provides, to the best of our knowledge, the first large sample test of the impact of the integration of cross-border clearing and settlement systems, a process whose benefits are currently under debate.³ Our finding that Global bonds' fungibility lowers borrowing costs provides evidence that the wave of consolidation and integration of securities clearing and settlement organizations and the call for the creation of a global clearing and settlement solution would indeed have benefits to firms' capital raising activities.⁴ Finally, our analysis of the shareholder wealth effects of Global, Euro and domestic bond issuance allows us to add to the literature that examines the wealth effects of security issuance by U.S. firms abroad (e.g. Kim and Stulz (1988), Chaplinsky and Ramchand (2000a)).

The paper is organized as follows: Section 2 describes the institutional features of Global bonds and their clearing and settlement procedures. Section 3 discusses potential benefits and costs of Global bonds relative to domestic bonds. Section 4 describes the Global and benchmark bond samples and provides some characteristics of the issuers. Section 5 compares the yield spreads and gross (underwriting) spreads of Global and comparison bonds to identify benefits to Global bond issuance. Section 6 analyzes the stock price reaction to the announcement of Global bond issuance. Section 7 discusses some of the additional robustness tests that were conducted. Section 8 concludes.

² See "The Rise of Global Bonds", Financial Market Trends, June 1994.

³ The only other security that shares the fungibility of global bonds is the Global Registered Share (GRS). A GRS is similar to an ordinary share except that investors can trade it on various stock exchanges around the world in many currencies. Currently, there are only 4 firms with GRS programs. For a detailed study of the first GRS, issued by DaimlerChrysler, see Karolyi (2003).

⁴ The *Depository Trust & Clearing Corporation* (DTCC), the world's largest securities clearing and settlement organizations, called for (October 2000) the financial services industry to develop a global clearing and settlement solution to lower costs, improve liquidity, and increase market capacity.

2. A Primer on Global bonds

Global bonds are relatively new financial instruments. In 1988 under the direction of the World Bank's Kenneth Lay, over 125 institutional investors in 16 countries were surveyed on how to improve debt issuance and trading.⁵ The results of this survey found that liquidity, investor base and trading convenience were the factors investors most valued. Given the large annual borrowing needs of the World Bank, they undertook the engineering of the Global bond, which entailed extensive legal work to integrate settlement procedures across systems. The first Global bond was issued by the World Bank in 1989, the first corporate Global bond was issued by Matsushita Electric in 1992, and the first U.S. firm's Global bond was issued by Walt Disney in 1996.⁶ Global bonds have several defining features. First, they are engineered to be fully fungible securities in that a Global bond can be traded in multiple markets without restrictions. Second, they are sold simultaneously in multiple markets, such as the U.S. and Euro market, at the same offer price. Third, they are extremely large offerings that are often offered in multiple tranches of differing size and maturity.

Since part of the issue is placed in the U.S., Global bond offers must be registered with the Securities and Exchange Commission (SEC).⁷ The marketing of a Global bond issue is similar to that of a domestic bond issue. A syndicate of underwriters, usually comprising of underwriters with a strong presence in the targeted market places, undertakes the marketing of the issue. Often, the syndicate manager for the international portion of a Global offer is the international affiliate of the domestic book manager. The issuer, often in consultation with the lead manager, appoints members of the underwriting syndicate. By law, the offer price must be the same for the domestic and non-U.S. tranches.⁸

⁵ See *Mobilizing private savings for development : IBRD and the capital markets* by Kenneth G. Lay, 1994. ⁶ Source: Securities Data Company

⁷ In all global issues with a U.S. component, the SEC had required that 100% of the issue be registered with the SEC. This was a precaution in case the entire issue was placed in the U.S. or the entire issue after initial placement flowed back into the U.S. However, the SEC made a decision in September 1993 to allow registration of only the portion placed in the U.S. and thus some margin for flow back from the Eurobond market. This could reduce the registration fees for global issues significantly and further lower the fixed cost of a global issue.

⁸ While geographic placement data are generally unavailable, we were able to gather this information for 18 global bonds and find, on average, that approximately 19 percent of the issue is placed outside the United States.

U.S. firms are the most prevalent issuers in the corporate Global bond market, and account for 70% of the issues.⁹ Appendix A provides information on a sample Global bond issue made by Wal-Mart. This issue comprised of three tranches issued on August 5th, 1999 that raised a total of 5.750 billion USD. One tranche was a two year bond that raised 1.250 billion USD, another was a 5 year bond that raised 1.250 billion USD, and the third tranche was a 10 year bond that raised 3.250 billion USD. Wal-Mart mentioned that almost the entire bond issue was placed with institutional investors in the U.S., Europe, Asia, and the Middle-East, with less than 1% placed with high net worth individuals. News articles relating to the bond issue mention the liquidity of the proposed bond issue as a key feature of the securities.¹⁰

2.1 Clearing and settlement procedures for Global bonds

Global bonds are book-entry bonds, which means that investors will not be entitled to receive physical delivery of the bonds in paper form. The book-entry system is used because it eliminates the need for physical movement of securities certificates and enables simultaneous electronic book-entry delivery against payment, thus eliminating the risk of trade-failures from the lack of simultaneous transfers of securities and cash between sellers and buyers.

Each Global bond is deposited with the Depository Trust Company (DTC) and is registered in the name of DTC or DTC's nominee. Purchasers of Global bonds in the U.S. may do so only through DTC, while purchasers of Global bonds in Europe can do so through Clearstream or Euroclear. Because DTC is the only registered owner of the bonds, Clearstream and Euroclear buy and sell Global bonds through their DTC depositaries, such as JP Morgan Chase.

Global bonds are designed to trade and settle like home market instruments from the investors' perspective. That is, Euro-market investors can trade and settle Global bonds like they would Eurobonds, while U.S. investors can trade and settle Global bonds like they would U.S. domestic bonds. For example, a Euro-market investor can use the same account with one of the European clearing houses (Clearstream or Euroclear) for settling transactions of both Global bonds and Eurobonds. If however, a Euro-market

⁹ Source: Securities Data Company

¹⁰ See "Wal-Mart to raise finance for Asda buy" in The Financial Times, July 20, 1999.

investor wants to trade in a pure U.S. domestic bond, he would have to open an account with a new broker/dealer who has an account with the DTC to execute the trade. To the extent that this lowers transactions cost for Global bonds relative to U.S. domestic bonds, Euro-market investors may place a higher value on Global bonds.

Cross-market trades (a U.S. investor trading with a Euro-market investor) in Global bonds can occur and the trading and settlement systems are set-up to handle such transactions. The U.S. investor conducts the transaction through his DTC depositary while the Euro-market investor uses the DTC depositary of Euroclear or Clearstream (JP Morgan Chase). The U.S. investor will receive credit for any bonds purchased or cash for any bonds sold, on the DTC settlement date. For Euro-market investors, because of time zone differences, the credits of Global bonds purchased or cash for any bonds sold will be received the business day following the DTC settlement date. Hence, another important advantage of Global bonds appears to be that cross-border transactions can occur more cost-efficiently relative to domestic bonds since the clearing and settlement systems are integrated for Global bonds.¹¹ Thus, the integration of clearing and settlement systems accentuates the benefits of fungibility of Global bonds.

3. Potential benefits and costs of globally fungible securities

Previous research has identified various market imperfections, such as information asymmetries, illiquidity, and transactions costs, which exist in financial markets. Issuers and investors can potentially benefit if fully fungible securities, like Global bonds, can partly overcome some of these imperfections. In this section, we highlight some of the potential benefits and costs of Global bonds relative to domestic bonds.

3.1 Liquidity

Global bonds have several features that may increase their liquidity relative to other types of debt instruments. First, they are extremely large offerings and liquidity is often found to be increasing in offer size (Hong and Warga (2000), Fisher (1959)). Second, Global bonds are targeted to a globally diverse investor base, which may result

¹¹ The main benefit to Euro-market and U.S. investors is that they can use their existing accounts with Euroclear, Clearstream and DTC to undertake cross-border transactions.

in increased divergence of opinion and more incentives to trade (Chen, Hong, Stein (2002), Scheinkman and Xiong (2002)). Finally, Global bonds also have longer trading hours since they can be typically traded in several international markets in different time zones. Therefore, a U.S. investor can sell his bonds in a European market even before regular trading hours in the U.S. market. Longer trading hours would seem to be a desirable feature for investors since they allow a timing option of being able to trade a bond almost around-the-clock. Werner and Kleidon (1996) and Chan, Fong, Kho and Stulz (1996) find that in the case of cross-listed equities, the arrival of information throughout non-trading periods can influence volatility, volume and bid-ask spreads. In fact, the World Bank states that Global bonds offer the opportunity to trade "…large volumes in any time zone at tight bid-offer spreads".¹² Therefore, if Global bonds are more liquid than domestic bonds, investors may be willing to pay a higher price, all things equal (Amihud and Mendelson (1986, 1991)).

3.1.1 Cross-border transactions cost

Cross-border trading in Global bonds is enhanced by the integration of clearing and settlement systems in the Euro-market (Clearstream and Euroclear) and U.S. market (DTC). Hence, international investors are likely to find cross-border transactions less costly, in terms of lower brokerage fees and quicker clearing and settlement, for Global bonds relative to domestic bonds.

3.2 Investor base and market conditions

Due to the potential benefits of higher secondary market liquidity, lower crossborder transaction costs and longer trading hours, international investors are likely to find fungible securities attractive. Hence, Global bonds may be an effective way to widen a firm's investor base, which can lower its cost of capital (Merton (1987)). In addition, global issuers can adjust each market's allocation based on investors' tastes and preferences in order to take advantage of their differential demand functions (Stulz and Wasserfallen (1995)). The ability to take advantage of market conditions across countries may be especially valuable for Global bond offers, since they are typically very large issues, which could make placing them in just the domestic market difficult.

¹² www.worldbank.org/debtsecurities

3.3 Issuing costs (gross spreads)

The issuing costs (gross spreads) in different bond markets may differ depending on factors like underwriter competition and economies of scale. Global bonds, given their large size, may induce greater competition between underwriters as well as offer economies of scale to underwriters. Hence, even if there are no other advantages to Global bonds like higher liquidity, issuers would prefer to issue Global bonds if it results in lower issuing costs since it lowers the issuers' cost of capital. High reputation underwriters are prominent players in the Global bond market, which is consistent with a positive relation between market share and reputation (Dunbar (2000)).

3.4 Costs of Global bond issuance

The fact that pure domestic issuance is still adopted by a large number of firms suggests that there are costs to global issuance that may offset some of the benefits. For example, a key determinant of selling costs in a foreign market is likely to be the visibility and reputation of an issuer.¹³ On the other hand, a firm without any global presence may have to incur significant selling costs in order to attract foreign investors which would make a global issue less attractive. In untabulated results, we find that over 45% of U.S. Global bond issuers had previously issued Eurobonds. This suggests that global issuers tend to be firms that are likely to have lower selling costs in Euro-markets.

Incremental fixed costs associated with a global issuance relative to a pure domestic issuance would include road shows and selling of the global issue in foreign markets, complying with the regulatory requirements of foreign markets, exchange listing requirements and costs associated with the linking of foreign market clearing houses (ClearStream and Euroclear) and the Depository Trust Company (DTC) to allow efficient clearing and settlement for cross-border trades of Global bonds.¹⁴ Therefore, the optimal size of a global issue may have to be large enough to generate financing cost savings (lower gross spreads and lower cost of capital) that offset these higher costs. The mean (median) Global bond in our sample is 0.914 billion USD (0.695 billion USD). This

¹³ The literature on Eurobonds suggests that traditionally Eurobond investors were attracted to issuer firms that were well known with significant overseas operations (see, e.g., Kidwell, Marr and Thompson (1985)).

¹⁴ Although no incremental information may be required relative to SEC requirements, the issuer would likely have to provide the information in a different format as required by the regulatory authorities in the foreign market. This may result in some additional administrative costs. Almost all global bonds are listed on the OTC (U.S.) and Luxembourg or London Exchanges.

observed large global issue size lends some support to the argument that there exist significant costs in the global issuance process.

4. Issue and Issuer Characteristics

4.1 Data selection and issue characteristics

The sample consists of all straight investment grade, fixed-rate coupon, U.S. dollar-denominated bond offerings, issued by non-financial firms in the Global bond market from January 1, 1996 to September 20, 2003. There are 230 Global bond issues in the sample, of which 179 were issued by U.S. firms and 51 issued by non-U.S. firms. In addition to these 230 issues, there was also one non-investment grade U.S. global issuer that was excluded.¹⁵

We benchmark the sample of U.S. firms' Global bonds in two ways. First, we compare the borrowing costs of Global bonds issued by U.S. companies to domestic bonds issued by U.S. companies. An advantage of this approach is that issuance procedures, flotation costs and indentures are uniform across Global and domestic bond issuance. A potential drawback of this benchmark is that while there are no restrictions to prevent foreign investors from holding domestic corporate bonds, the investor base is likely to be U.S. based, which could render fungibility less important. Therefore, we examine a second benchmark that consists of Eurobond issues by U.S. firms. Since both Eurobonds and Global bonds are targeted to non-U.S. investors, fungibility is likely to be the key difference in the bonds. The potential limitations of the Eurobond comparison sample are that there are relatively fewer Eurobonds issued in the sample period and that the relative yields could be influenced by differences in issuance procedures (registration and disclosure requirements tend to be more stringent and costly for Global bonds relative to Eurobonds), flotation costs (different gross spreads maybe charged by underwriters in different markets) and indentures (Eurobonds generally have fewer restrictive bond covenants than Global bonds).

To benchmark the Global bonds of non-U.S. firms, we gather data on bonds issued in the U.S. market by foreign firms (also known as Yankee bonds). Yankee bonds

¹⁵ The reason for excluding Federal-Mogul is to ensure a more homogenous sample in terms of credit quality since non-investment grade bonds forms a very small percentage of global issuance. In addition, it was announced around the date of issuance that Federal-Mogul was in financial distress.

are a natural benchmark since their characteristics are similar to Global bonds (e.g., registration, disclosure requirements, covenant provisions), target similar investors, but differ in fungibility since Yankees are placed solely in the U.S. public bond market.¹⁶

The data on Global bonds, Eurobonds, Yankee bonds and domestic bond issues is obtained from Securities Data Company (SDC) New Issues Database and Bloomberg. We only include fixed-coupon rate U.S. dollar-denominated offerings, which comprise the vast majority of global issues, to facilitate the comparison of borrowing costs. Finally, since the Global bond sample consists of only investment grade bonds, only investment grade comparison bonds are chosen to form the benchmark samples.

An interesting feature of Global bonds is that they are often issued in multiple tranches. That is, a firm issues Global bonds of different maturities on the same issue date. A potential reason for issuing multiple tranches could be to target different investor bases so as to make the sales of global issues less costly, since placing a large bond offering of a single maturity may be difficult. In our sample, for the issues by U.S. firms, 80 firms issued 1 bond on the same date (80 bonds), 28 firms issued 2 bonds on the same date (56 bonds), 13 firms issued 3 bonds on the same date (39), and 1 firm issued 4 bonds on the same date (4 bonds). For Global bonds from non-U.S. firms, there were 37 issues done in a single tranche (37 bonds), 4 issues done in double tranches (8 bonds) and 2 issues done in triple tranches (6 bonds). To facilitate comparison with domestic bonds in our analyses of yield and gross spreads, we treat each bond issue, even when it is part of a multi-tranche issuance, as a separate observation. This may result in lack of independence among some observations in the sample. This econometric issue and how it is addressed in this paper is discussed in Section 5.1.1.

Table 1 provides sample statistics for Global bonds and the benchmark domestic bonds, Eurobonds and Yankee bonds. As can be seen from the table, Global bond issuance has steadily increased since the beginning of the sample period. There are 179 Global bonds issued by U.S. firms and 51 Global bonds issued by non-U.S. firms. The U.S. domestic bond, Eurobond, and Yankee bond samples consists of 2231, 62, and 143 bonds, respectively. Examination of the issue characteristics of Global bonds and comparison bonds, however, suggests important differences. For example, global issues

¹⁶ See Miller and Puthenpurackal (2002) for an analysis of the Yankee bond market.

tend to be much larger than benchmark issues, and less frequently contain call provisions. In addition, we note differences in distributions of credit ratings, maturities and 1-digit sic industry groups (not tabulated) between the various samples. This highlights the need to control for issue characteristics in our analyses.

4.2 Characteristics of Global and non-Global Issuers

Kim and Stulz (1988) show that Eurobond issuers tend to be large, well-known firms with a global presence. They tend to have products sold world-wide, have strong brand names and have production plants in many countries. This suggests that Eurobond issuers are those with lower foreign selling costs. Since Global bonds are partly placed in the Eurobond market, one may expect that similar firms issue Eurobonds and Global bonds. Also, one would expect that Global bond issuers have large debt capacities since Global bonds tend to be large. Consistent with this notion, the typical Global bond issuers are well known companies and about 45% of global issuers had a prior Eurobond offering. Also, about 48% of global issuers have at least 1 foreign exchange listing. To the extent these firms are already recognized by international investors, global issuers are likely to have lower selling costs in foreign markets than their pure domestic counterparts.

Table 2 provides data on Global and benchmark bond issuers' characteristics such as total assets, market value of equity, leverage, profitability, Q-ratio and interest coverage. This financial data is obtained from Compustat and Worldscope and the items reported are for the latest financial year prior to the issue date. Comparing U.S. firms that issue Global bonds to U.S. firms that issue pure domestic bonds, based on medians, U.S. global issuers have about four times the assets and about six times the market value of equity of U.S. domestic issuers. The median global issue size (\$ 699 mill.) is over 3 times the median domestic issue size (\$ 198 mill.).¹⁷ However, as a proportion of market value of equity, the median global offer is smaller than the median domestic offer. Consistent with being high quality firms, global issuers have higher q-ratios. The median leverage ratios of global and domestic issuers are similar although the median leverage ratios of global issuers are lower. The dividend yields of global issuers are

¹⁷ The samples in Table 2 are smaller than those in Table 1 since some observations are lost during the merging of issue data with Compustat data. The characteristics of the samples in Table 1 and Table 2 are, however, similar.

significantly lower than that of domestic issuers while the median coverage ratio is higher for global issuers. While Table 2 also shows that the characteristics of U.S. firms that issue Global bonds are more similar to U.S. firms that issue Eurobonds, the average Global bond issuing firm remains larger than its Eurobond issuing counterpart.

Comparing non-U.S. firms that issue global bonds to non-U.S. firms that issue Yankee bonds, again we see that Global bond issuers are larger in size, issue proceeds, and Q-ratio. In our analyses, we treat different bond issues on the same date by a firm as separate observations. To check whether this is influencing the reported sample statistics, we recalculate (not tabled) the sample statistics after treating all bond issues by a firm on the same date as part of the same issue and find similar results.

The characteristics reported in Table 2 show that the typical global issuer and offer are much larger than the typical domestic issuer and offer. This suggests the need to control for differences in the issuer as well as issue characteristics of the two samples in our analyses. In addition, it raises the possibility that the decision to issue globally may be endogenously determined.

5. Testing for Benefits of Global Bonds

Fungible securities like Global bonds can potentially command a higher price than comparable domestic bonds due to higher liquidity, lower gross spreads, lower cross-border transactions costs and a wider investor base. This would predict that firms can lower their cost of capital by issuing fungible securities like Global bonds, ceteris paribus.

5.1 Comparing yields of Global bonds and domestic bonds issued by U.S. firms

To investigate whether global issuance lowers the cost of debt, our first set of tests examines the differences in at-issue yield spreads using a pooled sample of Global bonds and domestic bonds issued by U.S. firms. To do so, however, a few econometric issues need to be addressed. First, there is the lack of independence between bonds that are issued by the same firm on the same date. Second, the decision to issue globally may be endogenously determined by issuer and issue related characteristics, such as issuer and issue size.

5.1.1 Controlling for potential lack of independence of bond issues in multi-tranche issuances

Estimation using OLS assumes that observations are uncorrelated. If some observations are correlated due to the treatment of multi-tranche issues as independent issues, OLS is still unbiased and consistent. However, the standard errors of OLS are no longer correct. To control for this, in all our analyses, we adopt a variation of the standard robust estimator of variance (Huber (1967), White (1980)), developed by Rogers (1993), to compute robust standard errors.¹⁸ This procedure takes into account the possibility that observations within clusters may not be independent. We specify that bond issues by an issuer on the same date are part of the same cluster. ¹⁹

5.1.2 Controlling for potential endogeneity in the decision to issue globally

Since there are significant differences in the characteristics of global issuers and issues relative to domestic issuers and issues, it is possible that the decision to issue globally is endogenously determined. That is, global issuers may be a non-random sample of issuers that choose to issue globally because it is beneficial to do so. In the presence of endogeneity, results obtained using OLS, assuming an exogenous global dummy variable, are biased.

The econometric problem faced here is similar to the treatment effects model that considers the effect of an endogenously chosen binary treatment on another endogenous continuous variable, conditional on two sets of independent variables. We follow Maddala (1983) who derives the maximum likelihood estimator for the treatment effects model.²⁰

¹⁸ As an alternative to this robust variance estimator, we also employed a random firm effects estimation procedure and obtained similar results.

¹⁹ We also formed clusters based on issuer. That is, all issues by the same firm are considered part of the same cluster. The results using this cluster classification were similar to that using the cluster classification based on issuer-issue date. We also created a dummy variable that was equal to 1 when an observation was part of a multi-tranche offer and 0 otherwise. The results were similar when this dummy variable was included in the yield and gross spread analyses. Further details on the calculation of the robust estimator of variance are available from the authors upon request.

²⁰ See Reese and Weisbach (2002) and Doidge, Karolyi and Stulz (2004) for an application of the treatment effects model for firms cross listing in the United States.

5.1.3 Treatment effects model using maximum likelihood estimator

The primary regression equation of interest is

$$y_j = \beta x_j + \delta z_j + \varepsilon_j \tag{1}$$

where y_j correspond to yield spreads or gross spreads, x_j correspond to all the regressors used in the bond pricing model, and z_j corresponds to the global dummy variable. The binary variable z_j is assumed to stem from an unobservable latent variable

$$z_j^* = \mathcal{W}_j + u_j \tag{2}$$

where w_i are considered potential determinants of the decision to issue globally.

The decision to issue globally is made according to the rule

$$z_{j} = \begin{cases} 1, if \ z_{j}^{*} > 0 \\ 0, otherwise \end{cases}$$

where \mathcal{E} and u are bi-variate normal with mean zero and covariance matrix

$$\begin{bmatrix} \sigma & \rho \\ \rho & 1 \end{bmatrix}$$

The direction of the bias of the OLS estimates of the coefficient of the endogenous binary variable z_j , would depend on the sign of ρ , the covariance between ε and u. If ρ is positive, the OLS coefficient of the global dummy will be biased upwards. If ρ is negative, the OLS coefficient of the global dummy will be biased downwards.

The treatment effects model consists of a regression model and a treatment (selection) equation, and these are jointly estimated using full-information maximum likelihood. The use of a treatment effects model with a robust variance estimator allows us to control for potential endogeneity in the decision to issue globally and the lack of independence between bonds of a firm issued on the same date. We implement this model using the entire pooled sample of investment grade Global and domestic bonds

issued by non-financial U.S. firms, from January 1996 to September 2003, after merging with financial data from Compustat.

To test whether Global bond issuance lowers the cost of debt, we adapt the pricing model specification used by a number of previous studies to our treatment effects model. Studies that use this specification include Ederington (1975), Kidwell et al. (1985), Blackwell and Kidwell (1988), and Chaplinsky and Ramchand (2000b). These studies suggest that the yield on new issues of public debt is determined by default risk, the maturity of the issue, issue size, the presence of call provisions, and general economic conditions at the time of the sale. We examine the impact of global issuance on borrowing costs using multiple regression models that employ at-issue yield spread as the dependent variable. Our measure of the yield spread, YLDSPD, is defined as the offering yield-to-maturity (on the proceeds of the offer, after deducting total managers' fees) in excess of the yield on same-maturity treasuries.²¹ When a treasury bond of same maturity is not available, we interpolate between the two closest maturity treasury matches. The test variable of interest is the Global bond dummy variable. Robust standard errors are calculated after allowing for possible lack of independence between issues on the same date by a firm in all our analyses. The treatment effects model's regression and treatment (selection) equations' specifications are given below.

Regression Model

$$\begin{split} &YLDSPD = \beta_0 + \beta_1 GLOBAL + \beta_2 I_{(Aaa)} + \beta_3 I_{(Aa1)} + \beta_4 I_{(Aa2)} +\beta_{10} I_{(Baa2)} + \beta_{11} I_{(Baa3)} \\ &+ \beta_{12} LN(MATURITY) + \beta_{13} LN(PROCEEDS) + \beta_{14} RISKPREMIUM \\ &+ \beta_{15} CALLPROVISION + \beta_{16} SUBSIDIARY \\ &+ \beta_{17} Year 1996 + \beta_{18} Year 1997 \beta_{22} Year 2001 + \beta_{23} SIC1... + \beta_{29} SIC8 \\ &+ \beta_{30} \lambda \end{split}$$

²¹ We also constructed yield spread variables using yield on same *duration* treasuries instead of yield on *same* maturity treasuries. The results were unchanged.

Treatment Equation Model

$$GLOBAL = \alpha_0 + \alpha_1 ASSETS + \alpha_2 LN(PROCEEDS) + \alpha_3 (PROCEEDS/ASSETS)$$

+ $\alpha_4 (DEBT/ASSETS) + \alpha_5 COVERAGE + \alpha_6 ROA + \alpha_7 QRATIO$
+ $\alpha_9 DIVYIELD + \alpha_9 ISSUYEAR$

(4)

The control variables for the regression model account for differences in credit rating, maturity of issue, size of the issue, market risk premium, whether the issue has a call provision, year of issuance, industry of issuer and the subsidiary status of the firm. Because these variables have been used in previous studies, only a limited discussion is provided. We expect to find that the yield spread is negatively related to the quality of bond rating. The maturity of the issue is included to control for any term structure effects in the default premium. The size of the issue may be important if larger offerings have more public information than smaller issues, and therefore have less uncertainty. Also, large offerings may enhance future liquidity and hence may have lower yields. The variable RISK PREMIUM is defined as the yield spread between the Moody's Aaa seasoned corporate bond yield index and the composite Treasury yield index and is included to control for general economic conditions at the time of the sale. From the bondholder's perspective, bonds that are callable have prepayment risk. Therefore, we expect that callable bonds will have higher yield spreads. Some bonds are issued by subsidiaries of industrial companies with part of the proceeds being passed onto the parent. Since not all these bond issues are explicitly guaranteed by the parent, the market may demand higher yields from a bond issued by a subsidiary than if it were issued by the parent directly. To control for this possibility, we add a SUBSIDIARY dummy variable and predict a positive sign for its coefficient. The year dummies are included to control for general time effects in yields. The 1-digit SIC dummies are included to control for industry effects in yields.

The treatment equation attempts to control for the propensity of certain firms to issue Global rather than domestic bonds. We therefore include firm characteristic variables that were shown earlier to differ across the global and domestic sample. These control variables for the treatment equation account for issuer size (ASSETS), bond issue size (PROCEEDS), size of bond issue relative to issuer size (PROCEEDS/ASSETS), leverage related variables (DEBT/ASSETS and COVERAGE), profitability or quality variables (ROA and Q-Ratio), growth stage of issuer (DIVYIELDS) and year of issue (ISSUYEAR). The results for the treatment effects model are similar when market value related size and leverage measures are added to the treatment equation.

5.1.4 Relative borrowing costs of Global and domestic bonds of U.S. firms

The results of the estimated treatment effects model are reported in Table 3. Table 3 Probit I reports estimates for the treatment equation used to correct for selection bias. We note that the variables often have predicted signs and are significant. For example, we find that firm size and issue size tend to be positively correlated with the decision to issue globally which is consistent with the univariate analyses in Table 2.

The regression coefficient estimates of the treatment effects model, reported in Table 3 Model I, suggest that the borrowing costs on Global bonds is 20.7 basis points lower than domestic bonds, ceteris paribus. This provides direct evidence that Global bonds lower the cost of debt capital for issuers. We are able to interpret lower at-issue yields to imply lower cost of debt since the non-interest costs, such as restrictiveness of bond covenants, of Global bonds are similar to that of domestic bonds. Thus, firms that issue Global bonds, a fully fungible instrument, appear to obtain statistically and economically significant savings in their cost of capital. It also provides (indirect) evidence that investors value Global bonds because of their higher liquidity, lower crossborder transactions costs and longer trading hours. Further, it is interesting to note that our findings are consistent with market participants' perception of the economic impact of Global bonds. For example, in Euromoney (December 1993) the World Bank's Kenneth Lay states that before using the Global bond structure, the World Bank was paying 10 basis points over U.S. agencies for straight debt. Using the Global bond structure, the World Bank is paying around 8 basis points lower than U.S. agencies. He also cites liquidity and a wider and more diverse investor base as reasons for this savings.

Our finding that global security offerings lower the cost of capital also contributes to the literature that find a positive (or less negative) stock price reaction to international security offerings (Chaplinsky and Ramchand (2000a), Miller (1999), Foerster and Karolyi (1999), Errunza and Miller (2000), Foerster and Karolyi (2000), Kim and Stulz (1988)). Our result suggests that the documented positive stock price reaction could be, at least in part, driven by a lowering of the cost of capital rather than pure higher future cash flow effects.

5.1.5 Relative issuing costs (gross spreads) of Global and domestic bonds of U.S. firms

We examine the issuing costs (gross spreads) of Global bonds relative to domestic bonds for two reasons. One, it will help identify a potential benefit of global issuance that may partly explain the lower borrowing costs on Global bonds. Second, the analysis of gross spread data of Global and domestic bond issuance will provide new evidence on the relative cost of corporate bond issuance in different markets. Gross spreads are the fees underwriters charge for selling the firms' bond issue. The underwriter prepares the prospectus, organizes road shows and sells the issuers' story to potential investors. While there are other costs relevant to a global issue such as country-specific licensing fees, capital requirements and other compliance costs, gross spreads are likely to represent a significant portion of the costs of making a global offer (Chaplinsky and Ramchand (2000a)). Hence, we use gross spread, defined as the compensation paid to the underwriter for selling the firm's bond issue, as a percentage of the capital raised, to compare the issuing costs of Global and domestic bonds. We use a treatment effects model with the specifications described in Section 5.1.3 using gross spread as the dependent variable.

Table 3 Model II shows the regression results. The gross spread for global issues is 0.16 % lower than for domestic issues, ceteris paribus. Hence, one of the explanations for the lower borrowing cost on Global bonds is that underwriters charge a lower gross spread for global issues. For example, on a \$ 1 billion offer, this amounts to savings of gross (underwriting) fees of \$1.6 million. This suggests that issuers obtain statistically and economically significant savings in issuing costs through the issuance of Global bonds. Potential reasons for the lower gross spreads of Global bonds include greater underwriter competition in Global bond issuance given their large size as well as economies of scale given the likely significant fixed costs in the underwriting process. It may be noted that for this analysis, we use the reported gross spread data. Since there is a practice of rebating in the Eurobond market wherein large investors are offered rebates on the offer price by underwriters, the reported gross spread may overstate the true gross

spread. Since Global bonds are placed in the U.S. and Eurobond markets, there is a possibility that rebating also exists for Global bonds. However, our evidence shows that reported gross spreads for Global bonds are lower than the reported gross spreads for domestic bonds, therefore it follows that the true global gross spreads are lower than the true domestic bond gross spread, even after considering the possibility of rebating. Hence, this result provides new evidence on relative true gross spreads of corporate bond issuance in the domestic and Global bond markets.

We also find (untabulated results) that yields to investors (i.e., the yields before subtracting the gross spreads) for Global bonds are lower than those for domestic bonds, ceteris paribus. Hence, the lower cost of borrowing on Global bonds appears to be driven by both lower issuing costs (gross spreads) and by investors willing to accept lower yields on Global bonds due to the benefits of fungible securities.

5.2 Robustness tests

In addition to the use of the treatment effects model with a robust variance estimator, we conduct a number of robustness tests that use alternate controls and benchmarks.

As an additional control for issuer related differences, we benchmark the Global bond issues of U.S. firms to their respective domestic bond issues using OLS. This approach attempts to control for any issuer related characteristics not captured in the bond pricing specification that may influence yields and gross spreads. Models I and II of Table 4 show that employing the domestic bonds of global issuers as benchmarks produces borrowing costs and gross spread results that are similar to that obtained from the treatment effects model.

We also benchmark the sample of Global bond issues by U.S. firms to Eurobond issues by U.S. firms. Table 2 shows that Global bond and Eurobond issuers share relatively similar characteristics compared to the domestic bond issuers. Further, 45% of firms that issued Global bonds had previously issued Eurobonds. However, one characteristics that Global and Eurobonds differ substantially on is offer size (medians of \$699 and \$299 million, respectively). Therefore, for comparison purposes, we exclude Eurobonds that are smaller than the smallest Global bond (\$100 million).²² Model III of

²² Comparing with only the Eurobonds issued by global bond issuing firms also yields similar results.

Table 4 suggests the borrowing costs on Global bonds is 42.5 basis points lower than Eurobonds, ceteris paribus. While the magnitude of the coefficient is more than twice than the estimate derived from benchmarking to U.S. domestic bonds, it is likely a result of the higher reported gross spreads for Eurobond issues. Evidence of this is found in Model IV which shows that the gross spreads of Global bonds are 0.46% lower than for Eurobonds. In untabulated results, we also find that after accounting for gross fees investors are still willing to accept a lower return on Global bonds (28.7 basis points). Therefore, when we compare the borrowing costs, gross spreads and yield to investors of Global bonds to Eurobonds, we find consistent results. That is, Global bonds have lower required rates of return than less fungible debt instruments.

Table 4 also presents the results of tests that examine Global bonds issued by non-U.S. firms benchmarked to bonds issued in the U.S. domestic market by non-U.S. firms (Yankee bonds). Model V shows that compared to Yankee bonds, the borrowing costs of Global bonds are 19.9 basis points lower. Although the results for this comparison are statistically less significant (p-value=0.06) and economically smaller than some of our earlier estimates, they are largely consistent with our previous findings. That is, we find that the Global bonds issued by non-U.S. firms also enjoy lower borrowing costs than comparable but less fungible bonds. Further, Model VI shows that the gross spreads of Global bonds are lower than Yankee bonds.

Finally, as an additional approach to control for issue size differences in the U.S. global and domestic samples, we exclude all domestic offers that are smaller than the smallest Global bond offer (\$100 million) and then use OLS.²³ In untabulated results, we find that, after controlling for the bond and issuer characteristics, the borrowing costs on Global bonds is 19.9 basis points lower than that of domestic bonds. The borrowing costs and gross spread results are all similar to those obtained from the treatment effects model.

Overall, our results suggest that firms are able to lower their borrowing and issuing costs (gross spreads) by issuing Global bonds. Our results are consistent across U.S. and non-U.S. issuers and are robust to the use of alternate benchmarks. Our estimate of borrowing cost savings would predict that for the global issue described in Appendix

²³ We obtain similar results using other cutoffs such as US\$ 150 million, US\$ 200 million and US\$ 250 million.

A, Wal-Mart saved \$ 38.4 million on a \$ 5750 million issue by issuing Global bonds instead of domestic bonds.²⁴

6. Stock Price Reaction Analyses

To document the change in shareholders' wealth associated with issuing globally traded securities, we measure the stock price reaction to the announcement of U.S. firms' Global bond offerings. We also provide two comparison event studies as benchmarks. The first is the announcement effect of U.S. firms' issuing exclusively in the Eurobond market, a benchmark that represents another international corporate bond offering but does not share the fungibility of Global bonds. This comparison allows us to provide some indirect evidence on whether the stock price reaction to Global bond issuance is explained by Kim and Stulz's (1988) clientele hypothesis in which firms can exploit temporary differences in the Eurodollar/U.S. interest rates.²⁵ The second is the announcement effect of domestic U.S. corporate bond issuance by U.S. Global bond issuance is an artifact of the firms and time period rather than arising from benefits associated with global issuance.

Global bond issuance is often done in different tranches on the same day and hence, the number of unique announcement dates is less than the number of Global bond issues. We used Lexis-Nexis and Bloomberg to obtain announcement dates and check for contaminating stories around that date. This procedure resulted in 72 Global bond announcement dates. Using a similar procedure, we obtain 36 Eurobond announcement dates and 114 domestic bond announcement dates.

To measure abnormal returns, we estimate a market model for each firm using daily returns. As a proxy for the market return, we use the CRSP Equally Weighted index. Abnormal returns are then averaged across firms to form the average abnormal return. Tests of significance are conducted using standardized abnormal returns (Brown and Warner (1985)). We report results for the 3-day window (-1 to +1).

²⁴ The cost saving from each issue (this was a 3 tranche issue) is calculated as the difference in proceeds to Wal-Mart had it offered a yield 13.09 basis points higher than the actual yield.

²⁵ Henderson, Jegadeesh and Weisbach (2003) also document market timing by firms using country level data.

The event study results are reported in Table 5. We find that the mean abnormal stock price reaction, CAR(-1,+1), to the announcement of global issuance is 1.02% (median 0.72%) and is statistically significant at the 5% level. The sign-rank test is also significant at the 5% level.²⁶ The stock price reaction to the announcement of domestic bonds issued by global issuers over the same time period, however, is not significantly different from zero, consistent with Eckbo (1986). We also find that the stock price reaction to the announcement of Eurobond issuance by U.S. firms is not significantly different from zero for the same time period, which is consistent with the later period results of Kim and Stulz (1988). In Table 5, we also report a difference in mean test of CAR (-1,+1) for the three samples. The CAR(-1,+1) for the Global bond sample is, as expected, larger than that of the domestic and Eurobond sample, and is significant at the 5% and 10% level, respectively.

Overall, our results suggest that Global bond issuance is associated with a significant increase in shareholder wealth. Further, the positive stock price reaction to the announcement of global issuance does not appear to be driven by global issuers exploiting temporary differences in the Eurodollar/U.S. interest rates, but by benefits associated with the fungibility of Global bonds. While these findings are only suggestive given the small sample size, we also performed an additional analysis in which we estimate the interest cost savings based on the U.S.-Eurodollar interest rate difference and use it as an explanatory variable for the Global bond issuance announcement returns.²⁷ Consistent with our previous results (and subject to the same caveat of the low power of our test), we do not find support for the clientele hypothesis in explaining the positive benefits to Global bond issuance over our sample period.

7. Additional Tests

A number of additional tests were conducted to examine the liquidity of Global bonds (untabulated). Unfortunately, data for analyzing the liquidity of corporate bonds is not nearly as complete as that for stocks. For example, complete trading volume data

²⁶ The results are similar using the market-adjusted and mean adjusted benchmarks.

²⁷ In untabulated tests, we also found that the increase in the number of shareholders (Merton (1987)) did not help explain the observed announcement effects. We thank Rene Stulz for suggesting these tests.

(depth dimension of liquidity) for corporate bonds are currently not available.²⁸ However, data are available on corporate bond transactions by insurance companies since these companies are required by law to report to the National Association of Insurance Companies (NAIC) their securities transactions on Schedule D filings. We obtain data on U.S. corporate bond transactions of insurance companies from Capital Access International for 1996-2001.²⁹ As measures of liquidity for our domestic and Global bond samples, we analyze bid-ask spreads taken from Bloomberg and the frequency of daily non-zero returns using the average of daily bid and ask price quotes available on Bloomberg (Lesmond et al (1999)). We also examine the frequency of corporate bond transactions by insurance companies in Global and domestic bonds using data from Capital Access International.

Using historical daily bid and ask price quotes from Bloomberg for all investment grade Global and domestic bonds issued after 1995 by U.S. non-financial firms, we find that for the 6-month period after the issue date, the Global bond sample (79 obs.) mean (median) BASpread, defined as ((Ask Price-Bid Price)*100)/((Ask Price+Bid Price)/2), is 0.32% (0.31%) or 32bp, while that of the domestic sample (472 obs.) is 0.41% (0.35%). Therefore, we find that the mean and median BASpread of the global sample is significantly lower than that of the domestic sample. We also construct a variable called FREQNONZERO, defined as the ratio of the number of daily non-zero return observations to the total number of daily return observations since the bond was issued. For the Global bond sample (79 obs), the mean (median) of FREQNONZERO is 91% (97%) non-zero return days while that of the domestic sample (466 obs.) is 67% (75%) non-zero return days. The difference in means and medians of the frequency of non zero return days for the Global and domestic bond samples is significant at the 1% level. This suggests, using the Lesmond et al (1999) measure, that Global bonds were more frequently traded which we interpret as having lower transaction costs than domestic bonds. We also examine the frequency of corporate bond transactions by insurance

²⁸ One exception is the Fixed Income Pricing System (FIPS) introduced by the Nasdaq Stock Market in 1994 that provides data on complete trading volume and prices for a list of actively-traded high yield (junk) bonds. Currently, about 55 bonds are part of this list (Alexander et al (2000)).

²⁹ Insurance companies tend to focus on investment grade bonds and hence examining insurance company transactions in our sample bonds, which are all investment grade, is likely to be representative of all transactions in our sample bonds (Hong and Warga (2000)).

companies in Global and domestic bonds using data from Capital Access International from 1996-2001. We calculate the number of transactions (FREQTRAN) in the 6-month period after issue date for each bond. For the Global bond sample (47 obs), the mean (median) of FREQTRAN is 81.68 (69) transactions per 6 months while that of the domestic sample (449 obs.) is 37.63 (33) transactions per 6 months. The difference in means and medians of FREQTRAN for the Global and domestic bond samples is significant at the 1% level. This result also suggests that Global bonds are more liquid than domestic bonds.

Our analyses of various liquidity measures like bid-ask spreads, frequency of non-zero return days and frequency of bond transactions by insurance companies indicate that Global bonds are more liquid and have lower transaction costs than domestic bonds. Hence, higher liquidity appears to be an attractive feature associated with fungible securities.³⁰

8. Conclusion

This paper examines the potential benefits of security fungibility by conducting the first comprehensive analysis of Global bonds. These are very large bond offerings placed simultaneously in the U.S. and Eurobond markets at the same price and are fully fungible in that the identical instrument trades within each market as well as across markets without restrictions. We examine potential benefits of these fungible securities by studying their impact on firms' cost of capital, issuing costs (gross spreads) and shareholder wealth.

Using a sample of 230 Global bond issues by 94 companies from the U.S. and abroad over the period 1996-2003, we find evidence that suggest that firms are able to lower their cost of (debt) capital by issuing these fungible securities, and that this benefit is associated with the increased liquidity and lower issuing costs (gross spreads) of these instruments. Specifically, our results suggest that firms that issue globally are able to lower their borrowing costs by approximately 20 basis points relative to non-Global

³⁰ We also examined if our proxies for liquidity were priced in at-issue yields. Consistent with the U.S.based results of Crabbe and Turner (1995), we do not find our liquidity proxies are related to the at-issue yield spreads. However, an important caveat is that this test suffers from a look-ahead bias since the proxies for liquidity are constructed using data after the issue date.

bonds, ceteris paribus. This result is robust to a number of tests that attempt to control for alternative benchmarks, potential endogeneity in the decision to issue globally as well as issue size and issuer related differences. Our results are consistent with the hypothesis that firms are able to lower their cost of (debt) capital by issuing fungible debt securities that are more liquid, have lower cross-border transactions costs, have longer trading hours and can access a wider investor base.

We also document that the stock price reaction to the announcement of Global bond issuance is positive and significant, while comparable domestic and Eurobond issues by U.S. firms over the same time period are associated with insignificant changes in shareholder wealth. These results suggest that the benefit of global issuance is not being driven by global issuers exploiting temporary differences in the Eurodollar-U.S. interest rate. Overall, our findings suggest that the issuance of globally tradeable securities is associated with significant benefits.

References

- Alexander, Gordon J., Amy K. Edwards, and Michael G. Ferri, 2000, The determinants of trading volume of high yield corporate bonds, Journal of Financial Markets 3, 177-204.
- Amihud, Yakov and Haim Mendelson, 1986, Asset Pricing and the bid-ask spread, Journal of Financial Economics 17, 223-249.
- Amihud, Yakov and Haim Mendelson, 1991, Liquidity, maturity and the yields on U.S. Treasury securities, The Journal of Finance 46, 1411-1425.
- Blackwell, D. and D. Kidwell, 1988, An investigation of cost differences between public sales and private placements of debt, Journal of Financial Economics 22, 253-278.
- Brown, S., and J. Warner, 1985, Using daily stock returns: The case of event studies, Journal of Financial Economics 14, 3-31.
- Chen, J., H. Hong and J. Stein, 2002, Breadth of ownership and stock returns, Working paper, Harvard University.
- Chan, K.C., Wai-Ming Fong, Bong-Chan Kho, and Rene M. Stulz, 1996, Information, trading and stock returns: Lessons from dually-listed securities, Journal of Banking and Finance 20, 1161-1187.
- Chaplinsky, Susan, and Latha Ramachandran, 2000a, The impact of global equity offerings, The Journal of Finance 55, 2767-2789.
- Chaplinsky, Susan, and Latha Ramachandran, 2000b, The Impact of SEC Rule 144A on Corporate Debt Issuance by Foreign Firms, forthcoming, The Journal of Business.
- Crabbe, Leland E. and Christopher M. Turner, 1995, Does the liquidity of a debt issue increase with its size? Evidence from the corporate bond and medium-term notes markets, The Journal of Finance, December, Vol. 50, No. 5 1719-1734.
- Doidge, Craig, G. Andrew Karolyi and Rene M. Stulz, 2004, Why are foreign firms listed in the U.S. worth more?, Journal of Financial Economics 71, 205-238.
- Dunbar, Craig, 2000, Factors affecting investment bank initial public offering market share, Journal of Financial Economics 55, 3-41.
- Eckbo, B. Espen, 1986, Valuation effects of corporate debt offerings, Journal of Financial Economics, vol. 119-151.

- Ederington, L., 1975, Uncertainty, competition, and costs in corporate bond underwriting, Journal of Financial Economics 2, 71-94.
- Errunza, Vihang and Darius P. Miller, 2000, Market segmentation and the cost of capital in international equity markets, Journal of Financial and Quantitative Analysis, December Vol. 35 No.4.
- Fisher, L., 1959, Determinants of Risk Premiums on Corporate Bonds, Journal of Political Economy, June Vol. 67, 217-237.
- Foerster, Stephen R. and G. Andrew Karolyi, 1999, The effects of market segmentation and investor recognition on asset prices: Evidence from foreign stocks listing in the U.S., The Journal of Finance, forthcoming.
- Foerster, Stephen R. and G. Andrew Karolyi, 2000, The long-run performance of global equity offerings, Journal of Financial and Quantitative Analysis, December, Vol. 35 No.4.
- Gagnon, Louis and G. Andrew Karolyi, 2003, Multi-Market Trading and Arbitrage, working paper, Ohio State University.
- Henderson, B., N. Jegadeesh and M. Weisbach, 2003, World Markets for Raising New Capital, Working paper, University of Illinois.
- Huber, P.J., 1967, The behavior of maximum likelihood estimates under non-standard conditions. In *Proceedings of the Fifth Berkeley Symposium in Mathematical Statistics and Probability*. University of California Press.
- Hong, Gwangheon and Arthur Warga, 2000, An empirical study of bond market transactions, Financial Analyst Journal, March/April.
- Karolyi, Andrew, 2003, DaimlerChrysler A G, the First Truly Global Share, The Journal of Corporate Finance 9, 409-430.
- Kidwell, David S., M. Wayne Marr, and G. Rodney Thompson, 1985, Eurodollar bonds: Alternative financing for U.S. companies, Financial Management, Winter, 18-27.
- Kim, Y.C. and R.M. Stulz, 1988, The Eurobond market and corporate financial policy: A test of the clientele hypothesis, Journal of Financial Economics 22, 189-205.
- Lesmond, D.A., Joseph P. Ogden, and Charles A. Trzcinka, 1999, A new estimate of transaction costs, Review of Financial Studies, Winter, Volume 12, 1113-1141.

- Maddala, G.S., 1983, Limited-dependent and qualitative variables in econometrics, Cambridge University Press, New York.
- Merton, Robert C., 1987, A simple model of capital market equilibrium with incomplete information, Journal of Finance 42, 483-510.
- Miller, Darius P., 1999, The market reaction to international cross-listings: Evidence from depositary receipts, Journal of Financial Economics 51, 103-123.
- Miller, Darius P., and John J. Puthenpurackal, 2002, The costs, determinants, and wealth effects of international capital raising: Evidence from public Yankee bonds, Journal of Financial Intermediation 11, 455-485.
- Reese, William, and Michael Weisbach, 2002, Protection of minority shareholder interests, cross-listing in the United States, and subsequent equity offerings, Journal of Financial Economics 66, 65-104.
- Rogers, W.H., 1993. sg 17, Regression standard errors in clustered samples, Stata Technical Bulletin Reprints vol. 3, 88-94.
- Scheinkman, J. and W. Xiong, 2002, Overconfidence and speculative bubbles, Working paper, Princeton University.
- Stulz, Rene M. and Walter Wasserfallen, 1995, Foreign equity investment restrictions, capital flight, and shareholder wealth maximization: Theory and evidence, Review of Financial Studies 8, 1019-1057.
- Werner, Ingrid M., and Allan W. Kleidon, 1996, U.K. and U.S. trading of British crosslisted stocks: An intraday analysis of market integration, Review of Financial Studies 2, 619-664.
- White, H., 1980, A heteroskedasticity-consistent covariance matrix estimator and a direct test for heteroskedasticity, Econometrica 48, 817-830.

APPENDIX A

Wal-Mart's Global Issue

Issuer Name: Wal-Mart

Issue Date: August 5, 1999

Bond Issue Details: Three Tranche Issuance

US \$ 1250 million 2 year bond US \$ 1250 million 5 year bond US \$ 3250 million 10 year bond

Purpose of Issue: The proceeds from the global issue is to be used for refinancing the short-term borrowing used for acquiring Asda, a U.K. retail chain.

Cited Advantages: The news articles relating to this story mention Wal-Mart's high name recognition and the liquidity of the proposed bond issue, as key features of securities, bond investors in different markets are interested in.

Placement of Bond: Information obtained from Wal-Mart indicates that the bonds were placed in the U.S., Europe, Asia and the Middle-East.

Investor Base: Wal-Mart mentions that almost the entire bond issue was placed with institutional investors (money managers, pension funds, banks/trust and insurance companies) with less than 1% placed with high net worth individuals.



TABLE 1

Descriptive statistics for bonds issued by U.S. and Non-U.S. firms by bond type The U.S. sample consists of investment grade, fixed rate coupon, U.S.\$ denominated Global, domestic and Eurobonds issued by non-financial U.S. firms in the U.S. market from 1/1/96 to 09/20/03. The Non-U.S. sample consists of investment grade, fixed rate coupon, U.S. \$ denominated, Global and Yankee bonds issued by non-financial Non-U.S. firms in the U.S. market from 1/1/96 to 09/20/03. Panel A presents the time distribution of bond issues and Panel B presents information on Moody's ratings of sample bonds with Aaa indicating the highest quality. Means and medians (in parentheses) for issue size and years to maturity, and percentage of issues with call provisions are presented in Panel C.

	US Issuers		Non	-US Issuers	
	Global Bonds	Domestic Bonds	Eurobonds	Global Bonds	Yankee Bonds
Panel A: By offerin	g year				
	Number	Maaabaa	Number	Number	Normalian
1996	Number 1	Number 230	3	Number 3	Number 24
1990	3	333	3 14	1	24 29
1997	9	485	8	1 2	29 30
1998	9 19	280	8 9	4	12
2000	19 19	280 166	4	4	9
	19 50	273	4 3	4	9 24
2001				-	
2002	41	307	15	12	10 5
2003	37	157	6	14	5
Total	179	2231	62	51	143
Panel B: Rating Dis	stribution				
	Number	Number	Number	Number	Number
Aaa	5	47	3	1	-
Aal	-	24	4	5	-
Aa2	14	37	3	3	_
Aa3	23	91	5	2	_
A1	21	239	12	3	8
A2	38	365	7	15	17
A3	27	377	2	7	14
Baal	16	427	27	6	36
Baa2	25	389	, 17	2	49
Baa3	10	235	2	7	19
Total	179	2231	62	51	143
Panal Ct Summan	Statistics				
Panel C: Summary	Statistics				
Issue Size (\$ mil.)	915	206.57	405.48	929.3	344.6
	(700)	(150)	(300)	(600)	(250)
Years to Maturity	10.79	12.28	10.06	10.82	15.91
5	(9.97)	(9.99)	(5.09)	(10.00)	(10.02)
% of Issues with	35.8%	60.0%	67.7%	47.1 %	49.0 %
Call Provisions					

TABLE 2

Select financial characteristics of U.S. firms issuing Global, domestic and Eurobonds and non-U.S. firms issuing Global and Yankee bonds

Select financial characteristics are reported for non-financial U.S. firms that issue Global, domestic and Eurobonds and for non-U.S. firms that issue Global and Yankee bonds, that are U.S. \$ denominated, investment grade, fixed rate coupon bonds, from 1/1/1996 to 9/20/2003. The data is from Compustat and Worldscope and the items reported are for the latest financial year prior to the issue date after removing extreme values. The first row gives the means, the second row gives medians while the third row gives the number of observations in parentheses. ASSETS is total assets; PROCEEDS is the proceeds from the bond issue; MVE is market value of equity; PROCEEDS/MVE is proceeds from the issue divided by market value of equity; PROCEEDS/ASSETS is proceeds from the issue divided by ASSETS; Q-RATIO is (long term debt + debt in current liabilities + liquidating value of preferred stock + market value of equity) / ASSETS; DIV_YLD is the annual dividends paid / market value of equity; MKTLEV is (long term debt + debt in current liabilities) / book value of equity; DEBT_ASSETS is (long term debt + debt in current liabilities) / ASSETS; COVERAGE is operating income before depreciation divided by market value of equity; PROCEEDS / ASSETS; COVERAGE is operating income before depreciation divided by interest expense; YFMAT is the years to maturity of the bond issue.

		U.S. issuers	<u>s</u>	Noi	n-U.S. issuers
Variable	Global	Domestic	Eurobond	Global	Yankee
ASSETS (\$ mill.)	58683.2	13130.1	27174.5	75364.8	9264.5
	31691.0	6955.0	18293.0	42285.7	5676.6
	(179)	(1659)	(54)	(44)	(123)
PROCEEDS (\$ mill.)	910.4	222.3	404.0	926.6	343.1
	699.0	198.8	299.8	596.6	249.8
	(179)	(1659)	(62)	(51)	(143)
MVE (\$ mill.)	62375.8	14704.3	46481.1	72721.1	9286.4
	37810.6	6065.5	27539.9	37478.4	4246.3
	(178)	(1659)	(54)	(44)	(44)
PROCEEDS /MVE (%)	3.71	4.79	2.45	3.18	8.40
	1.85	2.61	1.20	2.30	6.56
	(178)	(1659)	(54)	(44)	(123)
PROCEEDS /ASSETS (%)	3.43	3.93	2.20	4.46	6.87
	2.24	2.25	1.12	2.90	5.31
	(179)	(1659)	(54)	(44)	(123)
ROA (%)	14.79	15.49	16.91	13.65	14.38
	14.47	15.00	16.27	11.38	13.55
	(179)	(1659)	(54)	(43)	(121)
Q-RATIO	1.945	1.535	2.165	1.714	1.369
-	1.374	1.167	1.639	1.229	1.137
	(178)	(1659)	(54)	(43)	(122)
DIV_YLD (%)	1.74	2.17	2.03	2.04	2.13
_ 、 ,	1.34	1.72	2.01	1.93	1.36
	(178)	(1659)	(54)	(40)	(118)
MKTLEV	0.621	0.498	0.393	0.343	0.464
	0.238	0.348	0.224	0.198	0.367
	(178)	(1659)	(54)	(43)	(123)
BOOKLEV	0.750	1.100	1.021	0.675	0.877
	0.640	0.986	0.954	0.697	0.759
	(174)	(1659)	(54)	(43)	(123)
DEBT_ASSETS	0.299	0.325	0.305	0.316	0.336
—	0.273	0.325	0.308	0.277	0.307
	(179)	(1659)	(54)	(43)	(123)
COVERAGE	11.61	9.65	12.87	8.575	7.060
-	8.11	6.74	9.28	7.279	6.101
	(179)	(1659)	(54)	(43)	(119)
YFMAT (years)	10.79	12.74	10.06	10.82	15.91
- (),	9.97	10.00	5.09	10.00	10.02
	9.97 (179)	10.00 (1659)	5.09 (62)	10.00 (51)	10.02 (143)

TABLE 3

Multivariate tests for Global issuer effects in borrowing costs and gross spreads, controlling for endogeneity in the decision to issue globally

Regression estimates of yield spreads and gross spreads on bond characteristics, market conditions and the Global bond test variable. The treatment effects model, using full maximum likelihood estimation, is used to ensure consistent estimates, in the presence of endogeneity in the decision to issue globally. The treatment effects model consists of a regression model and a treatment (selection) model that are jointly estimated. The sample consists of investment grade domestic and Global, fixed rate coupon, U.S.\$ bonds issued by non-financial U.S. firms, in the U.S. market from 1/1/96 to 09/20/03, after merging with financial data from Compustat and removing observations with outlier values. In Model 1, the left-hand side (LHS) variable is the yield-to-maturity (on the net proceeds of the offer, after deducting total managers' fees) in excess of the yield on similar maturity treasuries. This measures borrowing costs to issuers. In Model II, the LHS variable is the gross spread, measured as the difference between the offered amount and the proceeds to the issuer, expressed as a percentage of the offered amount. The right hand side (RHS) variables used in all the models are listed below. LN(MATURITY) and LN(PROCEEDS) are the natural logarithms of years to maturity and proceeds from the issue, respectively. The variable RISK PREMIUM is defined as the yield spread between the Moody's Aaa seasoned corporate bond yield index and the composite Treasury yield index and is included to control for general economic conditions at the time of the offer. CALL is a dummy variable indicating whether the bond is callable. SUBSIDIARY is a dummy variable that indicates that the issuing firm is a subsidiary of a public firm. LAMBDA is the Inverse Mills Ratio used to correct for sample selection bias. In all regressions, individual rating dummies for the Moody's rating of the bond issue, single digit sic dummy variables for the 1-digit SIC code of the issuer, and year dummy variables for the year of the bond issue are included but not reported. For example, $I_{(Aaa)}$ is equal to 1 if the bond issue is rated Aaa; 0 otherwise, SIC1 equal to 1 if the 1-digit SIC code of the issuer is 1; 0 otherwise, YEAR1996 is equal to 1 if bond issue was in 1996; 0 otherwise and so on. The treatment (selection) equation is a probit model with the GLOBAL dummy variable as LHS variable. GLOBAL takes value 1 for global issues and 0 for domestic issues. The RHS variables used for the treatment model are also listed below. ASSETS is total assets in millions of dollars; PROCEEDS/ASSETS is the proceeds from the issue divided by ASSETS; DEBT ASSETS is (long term debt + debt in current liabilities) / ASSETS; COVERAGE is operating income before depreciation divided by interest expense; ROA is operating income before depreciation divided by ASSETS; Q-RATIO is (long term debt + debt in current liabilities + liquidating value of preferred stock + market value of equity) / ASSETS; DIV_YLD is the annual dividends paid / market value of equity; ISSUYEAR is the year of bond issue. The reported fit of the selection equation is based on a cut-off equal to the proportion of Global bonds in the full sample. P-values (in parentheses) are computed using heteroskedastic consistent variance estimates that also take into account possible lack of independence between issues by the same firm on the same day. *, ** and *** indicate significance at the 10%, 5% and 1% levels, respectively.

	B	orrowing Cost	Gross St	oread
	Probit I	Model I	Probit II	Model II
GLOBAL		-0.207***		-0.1684***
GLODIL		(0.01)		(0.00)
LN(MATURITY)		0.0482**		0.1997***
		(0.01)		(0.00)
LN(PROCEEDS)		0.0088		-0.0086
		(0.46)		(0.11)
RISK PREMIUM		0.9488***		0.0658
		(0.00)		(0.12)
CALL		-0.0777**		-0.0182
		(0.02)		(0.22)
SUBSIDIARY		0.3258**		-0.0308
		(0.02)		(0.30)
LAMBDA		0.0107		0.0127
		(0.73)		(0.21)
ASSETS	4.8e-06**		4.3e-06*	
	(0.05)		(0.08)	
LN(PROCEEDS)	1.4255***		1.4309***	
	(0.00)		(0.00)	
PROCEEDS/	0.400.0.0.0.0.0			
ASSETS	-9.4993**		-9.8606**	
	(0.03)		(0.01)	
DEBT/ASSETS	-1.6547**		-1.6388**	
	(0.03)		(0.03)	
COVERAGE	0.0005		0.0007	
ROA	(0.83) -0.0376		(0.74) 0.0210	
KUA				
Q-RATIO	(0.98) 0.1183		(0.99) 0.1211*	
Q-KATIO	(0.11)		(0.09)	
DIV YIELD	-5.1410		-5.2342	
DIV HELD	(0.27)		(0.26)	
ISSUEYEAR	0.2178***		0.2204***	
15501112/11	(0.00)		(0.00)	
RATINGS DUM	(0.00)	Included	(0.00)	Included
SIC DUM		Included		Included
YEAR DUM		Included		Included
INTERCEPT	-444.81***	-0.2137	-449.93***	0.1093*
	(0.00)	(0.14)	(0.00)	(0.08)
Observations	1832	1832	1832	1832
Predictive ability	0.86	1032	0.86	1032
of Probit Model	0.00		0.00	
Model Chi-Square		2197 (0.00)		1759
model em square		2177 (0.00)		(0.00)

ECB Working Paper Series No. 426 January 2005

38

			TABLE 4	•	•	
JLS regressions are u lomestic and Global, i sued by Global bond, inancial U.S. firms. T he U.S. market. For al jlobal bond are inclu, nanagers' fees) in exc neasured as the differ naticates that the issue arriable RISK PREMI ontrol for general ecol hat the issuing firm is lummy variables and y ake into account possil	OLS regressions are used to estimate the impact of Global bond issuance on borrowing costs and gross spreads. The sample for Models I and II consists of investment grade domestic and Global bond issued by non-financial U.S. firms, in the U.S. market with the domestic sample including only domestic bonds that we issued by flobal bond issuers. The sample for Models III and IV consists of investment grade, fixed rate coupon, USS denominated Eurobonds and Global bonds issued by non-U.S. firms in the U.S. market: For all samples, the sample for Models III and IV consists of investment grade, fixed rate coupon, USS denominated Eurobonds and Global bonds issued by non-U.S. firms in the U.S. market. For all samples, the sample for Models III and V, the left-hand side (LHS) variable is the offering yield-to-maturity (on the net proceeds of the offer, after deducting total managers' fees) in excess of the yield on same maturity treasures. This measures borrowing costs to issuers. In Models II, IV and VI, the LHS variable is the offering yield-to-maturity (on the net Proceeds of the offer, after deducting total measures that the issue is Global bond as the yield spread between the proceeds to the issue, expressed as a percentage of the offered amount. GLOBAL is a dummy variable that indicates that the issue is Global bond. LN(MATURITY) and LN(ROCEEDS) are the natural logarithms of years to maturity and proceeds from the issue, respectively. The variable RISK PREMIUM is defined as the yield spread between the Moody's Aaa seasoned corporate bond yield index and the composite Treasury yield index and is included to control for general economic conditions at the time of the offer. CALL is a dummy variable indicating that the bond is callable. SUBSIDIARY is a dummy variable that indicates that the issuing firm is a subsidiary of a public firm. PRIVPLAC is a dummy variable is a computed using the the bond is callable. SubSiDIARY is a dummy variable suit at the instrume variable suit of account possible lack of independence betw	EXAMPLATION OF COMPANIATION CONTRACT estimate the impact of Global bond issuance interaction of Company on the comparise of the comparise of the company of the sample for Models II and IV consists of investme less, the sample period is 1/1/96 to 09/20/03 a Models I, III and V, the left-hand side (LF the yield on same maturity treasuries. This retween the offered amount and the proceeds lobal bond. LN(MATURITY) and LN(PROC defined as the yield spread between the Mood- conditions at the time of the offer. CALL is a diary of a public firm. PRIVPLAC is a dummy mmy variables are included but are not reporte OIS firms: Globals vs Domestic bonds Borrowing Cost (II) Gross Shread	ate the impact of Global bond issuance on borrowing costs and gross spreads. The sample for Models I a upon, U.S.\$ bonds issued by non-financial U.S. firms, in the U.S. market with the domestic sample includit e sample for Models II and IV consists of investment grade, fixed rate coupon, US\$ denominated Eurobond; or Models V and VI consists of investment grade, fixed rate coupon, US\$ denominated Eurobond; or Models III and IV consists of investment grade, fixed rate coupon, US\$ denominated Eurobond; or Models V and VI consists of investment grade, fixed rate coupon, US\$ denominated Eurobond; or Models V and VI consists of investment grade, fixed rate coupon, US\$ denominated, Yankee and Global he sample period is 1/1966 to 09/20/03 and only domestic bonds, Eurobonds and Yankee bonds that are greadels I, III and V, the left-hand side (LHS) variable is the offering yield-to-maturity (on the net proceeds yield on same maturity treasuries. This measures borrowing costs to issuers. In Models II, IV and VI, the en the offered amount and the proceeds to the issuer, expressed as a percentage of the offered amount. GI (bond. LN(MATURITY) and LN(PROCEEDS) are the natural logarithms of years to maturity and proceeds as the yield on same maturity treasuries. This measures borrowing costs to issuers. In Models II, IV and VI, the en the offered amount and the proceeds to the issuer, expressed as a percentage of the offered amount. GI bond. LN(MATURITY) and LN(PROCEEDS) are the natural logarithms of years to maturity and proceeds ed as the yield on same between the Moody's Aaa seasoned corporate bond yield index and the composite Trea itions at the time of the offer. CALL is a dummy variable indicating that the bond is callable. SUBSIDIARY of a public firm. PRIVPLAC is a dummy variable indicating that were privately placed. In all regressions, individivations at the time of the offer. CALL is a dummy variable such ease day. *.** and *** indicate significance at the 10% trans; Globals vs Domestic bonds (TI) Gross Soread (TI) Gr	wing costs and gross spreads. The sample firms, in the U.S. market with the domest ment grade, fixed rate coupon, US\$ denominated, Yai domestic bonds, Eurobonds and Yankee bo ble is the offering yield-to-maturity (on t borrowing costs to issuers. In Models II, borrowing costs to issuers. In Models II, suer, expressed as a percentage of the off re the natural logarithms of years to matu easoned corporate bond yield index and th variable indicating that the bond is callable boonds that were privately placed. In all reg ues (in parentheses) are computed using he he same day. *, ** and *** indicate signif US firms: Globals vs Eurobonds	e for Models I and II cons e for Models I and II cons inated Eurobonds and Glol nkee and Global bonds issi nds that are greater or equ he net proceeds of the off IV and VI, the LHS vari fered amount. GLOBAL it rity and proceeds from the e composite Treasury yield rity and proceeds from the e composite Treasury yield ressions, individual rating teroskedastic consistent va icance at the 10%, 5% and Non-US firms: G (V) Borrowing Cost	odels I and II consists of investment grade le including only domestic bonds that were Eurobonds and Global bonds issued by non- d Global bonds issued by non-U.S. firms in t are greater or equal in size to the smallest proceeds of the offer, after deducting total J VI, the LHS variable is the gross spread, nount. GLOBAL is a dummy variable that I proceeds from the issue, respectively. The osite Treasury yield index and is included to IDIARY is a dummy variable that indicates s, individual rating dummies, single digit sic dastic consistent variance estimates that also t the 10%, 5% and 1% levels, respectively. Non-US firms: Globals vs Yankees
GLOBAL	-0.1309**	-0.1215***	-0.4253***	-0.4628***	-0.1996*	-0.1183***
	(0.03)	(0.00)	(0.00)	(0.00)	(0.06)	(00.0)
LN(MATURITY)	0.1934^{***}	0.2601^{***}	0.1861^{***}	0.1156^{***}	-0.0905	0.1727^{***}
I N/DDOCEEDS)	(0.00) 0 1078***	(0.00)	(0.00)	(0.00) 0.0668*	(0.15)	(0.00)
(KUCEEUS)	(0.00)	0.0294 (0.12)	(0.01)	-0.0000° (0.05)	(0.89)	0.0124 (0.35)
RISK PREMIUM	0.7306***	-0.0581	0.5191**	-0.0874	0.3001	-0.0132
CALL	(0.00) -0.0266	(0.36) 0.0008	(0.03) 0.0383	(0.48) -0.0319	(0.30)-0.1745	(0.81)-0.0483**
	(0.62)	(0.98)	(0.73)	(0.48)	(0.13)	(0.02)
SUBSIDIARY	0.1172	-0.0926*	-0.1636	-0.3620***	-0.0626	0.0203
PRIVPIAC	(0.33)	(0.08) -	(0.28) -0 3770	(0.00)	(0.57)	(0.28) -
			(0.15)	(0.71)		
INTERCEPT	-1.6800^{***}	-0.1955	-1.8352***	1.6456^{***}	1.0279	0.2261^{**}
	(0.00)	(0.27)	(0.00)	(0.00)	(0.17)	(0.03)
No. of observations \mathbb{R}^2	3/8 0.65	378 0.56	240 0 65	240 0 50	0.72	0.79
	0000	0	20:0	02:0	1.0	

39

TABLE 5 Stock Price Reaction to Global Bond, Eurobond and Domestic Bond Announcements

The samples consist of 72 announcements of Global bond offerings made by 40 U.S. firms, 114 announcements of domestic bond offerings made by 40 U.S. firms that have also issued Global bonds, and 36 announcements of Eurobond offerings made by 25 U.S. firms. The sample period for all the bond samples is 1996-2002. Stock price reactions for announcements in 2003 are not included due to non-availability of CRSP data for 2003. Abnormal returns are obtained using the market model and parameters are estimated over a 100 day period, from day -125 to -26 relative to the announcement date. The CRSP equally weighted index is used as a proxy for the market portfolio. Cumulative abnormal returns for the period -1 to +1, CAR (-1,1), for all the samples are presented with the announcement date as day 0. The z-statistic and sign rank z statistic are presented in parentheses below. *, ** and *** indicate significance of the z-statistic at the 10%, 5% and 1% levels, respectively. Tests of differences in mean CAR (-1,1), using the market model, between the different samples are also reported. P-values are in parentheses below.

	Global Bonds (1996-2002) (72 obs)	Domestic Bonds (1996-2002) (114 obs)	Eurobonds (1996-2002) (36 obs)	Test of difference between CAR (-1,1) of Global and domestic samples	Test of difference between CAR (-1,1) of Global and Eurobond samples
CAR (-1,1) Z–statistic Generalized Sign Rank Test Z stat.	1.02% ** (2.50) ** (2.06) **	0.04% (0.46) (0.26)	-0.38% (-0.90) (-1.51)	0.99% ** (0.04)	1.40% * (0.07)



European Central Bank working paper series

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.int)

- 395 "Fiscal sustainability and public debt in an endogenous growth model" by J. Fernández-Huertas Moraga and J.-P. Vidal, October 2004.
- 396 "The short-term impact of government budgets on prices: evidence from macroeconomic models" by J. Henry, P. Hernández de Cos and S. Momigliano, October 2004.
- 397 "Determinants of euro term structure of credit spreads" by A. Van Landschoot, October 2004.
- 398 "Mergers and acquisitions and bank performance in Europe: the role of strategic similarities" by Y. Altunbas and D. Marqués Ibáñez, October 2004.
- 399 "Sporadic manipulation in money markets with central bank standing facilities" by C. Ewerhart, N. Cassola, S. Ejerskov and N. Valla, October 2004.
- 400 "Cross-country differences in monetary policy transmission" by R.-P. Berben, A. Locarno, J. Morgan and J. Valles, October 2004.
- 401 "Foreign direct investment and international business cycle comovement" by W. J. Jansen and A. C. J. Stokman, October 2004.
- 402 "Forecasting euro area inflation using dynamic factor measures of underlying inflation" by G. Camba-Méndez and G. Kapetanios, November 2004.
- 403 "Financial market integration and loan competition: when is entry deregulation socially beneficial?" by L. Kaas, November 2004.
- 404 "An analysis of systemic risk in alternative securities settlement architectures" by G. Iori, November 2004.
- 405 "A joint econometric model of macroeconomic and term structure dynamics" by P. Hördahl,O. Tristani and D. Vestin, November 2004.
- 406 "Labour market reform and the sustainability of exchange rate pegs" by O. Castrén, T. Takalo and G. Wood, November 2004.
- 407 "Banking consolidation and small business lending" by E. Takáts, November 2004.
- 408 "The great inflation, limited asset markets participation and aggregate demand: FED policy was better than you think" by F. O. Bilbiie, November 2004.
- 409 "Currency mismatch, uncertainty and debt maturity structure" by M. Bussière, M. Fratzscher and W. Koeniger, November 2004.
- 410 "Do options-implied RND functions on G3 currencies move around the times of interventions on the JPY/USD exchange rate? by O. Castrén, November 2004.
- 411 "Fiscal discipline and the cost of public debt service: some estimates for OECD countries" by S. Ardagna, F. Caselli and T. Lane, November 2004.
- 412 "The real effects of money growth in dynamic general equilibrium" by L. Graham and D. J. Snower, November 2004.

- 413 "An empirical analysis of price setting behaviour in the Netherlands in the period 1998-2003 using micro data" by N. Jonker, C. Folkertsma and H. Blijenberg, November 2004.
- 414 "Inflation persistence in the European Union, the euro area, and the United States" by G. Gadzinski and F. Orlandi, November 2004.
- 415 "How persistent is disaggregate inflation? An analysis across EU15 countries and HICP sub-indices" by P. Lünnemann and T. Y. Mathä, November 2004.
- 416 "Price setting behaviour in Spain: stylised facts using consumer price micro data" by L. J. Álvarez and I. Hernando, November 2004.
- 417 "Staggered price contracts and inflation persistence: some general results" by K. Whelan, November 2004.
- 418 "Identifying the influences of nominal and real rigidities in aggregate price-setting behavior" by G. Coenen and A. T. Levin, November 2004.
- 419 "The design of fiscal rules and forms of governance in European Union countries" by M. Hallerberg, R. Strauch and J. von Hagen, December 2004.
- 420 "On prosperity and posterity: the need for fiscal discipline in a monetary union" by C. Detken,V. Gaspar and B. Winkler, December 2004.
- 421 "EU fiscal rules: issues and lessons from political economy" by L. Schuknecht, December 2004.
- 422 "What determines fiscal balances? An empirical investigation in determinants of changes in OECD budget balances" by M. Tujula and G. Wolswijk, December 2004.
- 423 "Price setting in France: new evidence from survey data" by C. Loupias and R. Ricart, December 2004.
- 424 "An empirical study of liquidity and information effects of order flow on exchange rates" by F. Breedon and P. Vitale, December 2004.
- 425 "Geographic versus industry diversification: constraints matter" by P. Ehling and S. B. Ramos, January 2005.
- 426 "Security fungibility and the cost of capital: evidence from global bonds" by D. P. Miller and J. J. Puthenpurackal, January 2005.